Author’s response to reviews

Title: Medial Open Transversus Abdominis Plane (Motap) Catheters For Analgesia Following Open Liver Resection: Study Protocol For A Randomized Controlled Trial

Authors:

Paul J Karanicolas (paul.karanicolas@sunnybrook.ca)
Sean Cleary (Sean.Cleary@uhn.ca)
Paul McHardy (paul.mchardy@sunnybrook.ca)
Stuart McCluskey (stuart.mccluskey@uhn.ca)
Jason Sawyer (jason.sawyer@sunnybrook.ca)
Salima Ladak (salima.ladak@uhn.ca)
Calvin Law (calvin.law@sunnybrook.ca)
Alice Wei (alice.wei@uhn.ca)
Natalie Coburn (natalie.coburn@sunnybrook.ca)
Raynauld Ko (raynauld.ko@uhn.ca)
Joel Katz (joeldkatz@gmail.com)
Alex Kiss (alex.kiss@ices.on.ca)
James Khan (james.khan@medportal.ca)
Coimbatore Srinivas (coimbatore.srinivas@uhn.ca)
Jenny Lam-McCulloch (jenny.lam-mcculloch@sunnybrook.ca)
Hance Clarke (hance.clarke@uhn.ca)

Version: 3 Date: 25 May 2014

Author’s response to reviews:

All formatting changes made as requested:

Please ensure the title conforms to journal style for study protocol articles. The title should follow the format “___________: study protocol for a randomized controlled trial.”

2. Please include the date of registration with your trial registration number at the end of your Abstract.

3. Please move your ethics statement to the Methods section and include the reference numbers given with ethical approval.

4. Please include a list of abbreviations used and their meanings, after your Trial Status.

5. Please mention each author individually in your Authors’ Contributions section. We suggest the following kind of format (please use initials to refer to each author’s contribution): “AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the
immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.”